\documentstyle[12pt]{article}
\input{psfig.sty}
\input epsf
\oddsidemargin 0cm      % this is margin in addition to the usual 1 inch
\evensidemargin 0cm
\textwidth 16.5cm

%\oddsidemargin -0.5cm      % this is margin in addition to the usual 1 inch
%\evensidemargin-0.5cm
%\topmargin -1.9cm
%\topmargin 0.0cm
%\baselineskip 0.0cm

%\textwidth 16.5cm
%\textwidth 17.0cm
%\textheight 24.7cm

\newcommand{\bi}{\begin{itemize}}
\newcommand{\ei}{\end{itemize}}
\newcommand{\be}{\begin{enumerate}}
\newcommand{\ee}{\end{enumerate}}
\newcommand{\tand}{\wedge}
\newcommand{\tor}{\vee}
\newcommand{\union}{\cup}
\newcommand{\intersection}{\cap}
\newcommand{\ch}{$\surd$}
\newcommand{\pgm}[1]{{\sc #1}}
\newcommand{\la}{\leftarrow}
\newcommand{\ra}{\rightarrow}

\begin{document}
\begin{center}
\large
{\bf Machine Learning 15-681/781}\\
Tom Mitchell, Fall 1998
\end{center}

\begin{center}
\large  Homework Assignment 1\\
Due at beginning of class on September 22, 1997 \\
\ \\

\end{center}

\be
\item Turn in a photograph of yourself (a photocopy of your student ID photo
will do --  we just want to know which face goes with which name). 
\item Exercise 2.3 in the textbook.
\item Exercise 2.4
\item Exercise 7.2, part {\em a} only.
\ee

\vspace*{1.5in}
\centerline{\b Reminder: course web site is www.cs.cmu.edu/afs/cs.cmu.edu/project/theo-3/www/ml.html}
\end{document}
